Author's response to reviews

Title: Angiotensinogen M235T gene variants and its association with essential hypertension and plasma renin activity in Malaysian subjects: A case control study

Authors:

Yee-How Say (sayh@hotmail.com)
King-Hwa Ling (khling@medic.upm.edu.my)
Gnanasothie Duraisamy (dural@medic.upm.edu.my)
Suzanne Isaac (rozita@medic.upm.edu.my)
Rozita Rosli (rozita@medic.upm.edu.my)

Version: Date: 30 March 2005

Author's response to reviews:

Dear Editorial Board,

All authors have read and agreed with all the changes made on the final version of the manuscript. Please find the following are descriptions for all the changes made on the manuscript.

Body of the Manuscript

Page 2

1. The last 2 sentences of the Background section were moved to Methods.

2. Additional of 2 sentences which are now leading the Background section of the abstract.

3. Additional of 'not only'... 'but also' into the last sentence of the Background section.

Page 4, Line 9 - 'alleleic' was changed to 'allelic'.

Page 4, Line 17 - 'AGT gene M235T variant' was changed to 'AGT M235T gene variant'.

Page 5, Line 9 - 'considred' was changed to 'consider'.

Page 5, Line 18 - Subheading 'BMI measurements' was changed to 'body mass index measurements'.

Page 5, Line 28 - Subheading 'Blood Collection' was changed to 'Blood collection'.

Page 7, Line 3 - 'gradient' was changed to 'Gradient'.

Page 7, Line 10 - Subheading 'Plasma Renin Activity Radioimmunoassay' was changed to 'Plasma renin activity radioimmunoassay'.

Page 7, Line 19 - 'coffecient' was changed to 'coefficient'.

Page 7, Line 22 - Subheading 'Statistical Analysis' was changed to 'Statistical analysis'.

Page 7, Line 28 - 'antropometric' was changed to 'anthropometric'.

Page 8, Line 12 - Subheading 'Demographic Data' was changed to 'Demographic data'.

Page 9, Line 2 - Subheading 'Genotypes and Alleles Frequencies' was changed to 'Genotypes and alleles frequencies'.
Page 10, Line 5 - Subheading 'Plasma Renin Activity (PRA)' was changed to 'PRA levels'.

Page 10, Line 17 - Subheading 'BP Variations' was changed to 'BP variations'.

Page 11, Line 8 - Subheading 'Associated Risk Factors' was changed to 'Associated risk factors'.

Page 11, Line 24 - 'hypertesnives' was changed to 'hypertensives'.

References - All authors' name for reference no. 5 and 23 were submitted.

Figures and table titles - Corrected as requested.

Figures - Figures were removed from the manuscript.

Tables - Table legends and titles were placed directly above the table that they represent. All text within the table were centered.

Thank you.